Author's response to reviews

Title: New Highly Active Antiretroviral drugs and generic drugs for the treatment of HIV infection: a budget impact analysis on the Italian National Health Service

Authors:

Umberto Restelli (urestelli@gmail.com)
Francesca Scolari (francesca.scolari2@gmail.com)
Paolo Bonfanti (pa.bonfanti@ospedale.lecco.it)
Davide Croce (d.croce@tiscali.it)
Giuliano Rizzardini (rizzardini.giuliano@hsacco.it)

Version: 2
Date: 11 February 2015

Author's response to reviews:

Dear editor,

as requested, lines and pages numbers has been added to the manuscript.
The "Author contribution" section was renamed "Acknowledgements" and the phrase "No funding was received for the study presented in the manuscript" has been added.

Kind regards,

Umberto Restelli